Author's response to reviews

Title: Estimating and validating disability-adjusted life years at the global level: a framework for cancer

Authors:

Isabelle Soerjomataram (soerjomatarami@iarc.fr)
Joannie Lortet-Tieulent (tieulentj@iarc.fr)
Jacques Ferlay (Ferlay.J@iarc.fr)
David Forman (formand@iarc.fr)
Colin Mathers (mathersc@who.int)
Donald Maxwell Parkin (max.parkin@ctsu.ox.ac.uk)
Freddie Bray (brayf@iarc.fr)

Version: 2 Date: 29 March 2012

Author's response to reviews: see over
Dear Sir/Madam,

**Estimating and validating disability-adjusted life years at the global level: a methodological framework for cancer**

Please find enclosed revised version of the above manuscript to be considered for publication as an original article.

We have adjusted our paper according to the reviewer comments and we think we have addressed all their comments. We believe that after these adjustments our paper has markedly improved.

We confirm that this paper is not submitted for publication elsewhere and the work done within does not duplication other work using the datasets. We further declare no conflict of interest.

We sincerely hope that this version of our paper will be suitable for publication in the BMC Medical Research Methodology.

On behalf all authors.
Yours sincerely,

Isabelle Soerjomataram